Title: Effectiveness of early discharge planning in acutely ill or injured hospitalized older adults: A systematic review and meta-analysis.

Authors:

mary t fox (maryfox@yorku.ca)
malini persaud (malinip@yorku.ca)
ilo maimets (ilo@yorku.ca)
dina brooks (dina.brooks@utoronto.ca)
kelly o'brien (kelly.obrien@utoronto.ca)
deborah tregunno (tregunno@yorku.ca)

Version: 2 Date: 26 April 2013

Author's response to reviews: see over